Author's response to reviews

Title: Immunophenotypic studies of monoclonal gammopathy of undetermined significance

Authors:

Horatiu Olteanu (holteanu@mcw.edu)
Huan-You Wang (Huan-You.Wang@UTSouthwestern.edu)
Weina Chen (Weina.Chen@UTSouthwestern.edu)
Robert W McKenna (mcken138@umn.edu)
Nitin J Karandikar (Nitin.Karandikar@UTSouthwestern.edu)

Version: 4 Date: 26 November 2008

Author's response to reviews:

Reply to requested formatting changes 11-26-08

Major revisions
The author’s contribution paragraph was changed by replacing the full names with initials and adding the sentence that specifies all authors have read and approved the final version of the manuscript.

Minor revisions
The title was formatted to sentence case.
Figures was changed to Figure legends.
The manuscript was re-checked for typographical errors and none were found.

Regards,
Horatiu Olteanu, MD, PhD